Author's response to reviews

Title: Body fat measurement by bioelectrical impedance versus air displacement plethysmography: a cross-validation study to design bioelectrical impedance equations in Mexican adults

Authors:

Nayeli Macias (nmacias@insp.mx)
Heliodoro Aleman-Mateo (helio@cascabel.ciad.mx)
Julian Esparza-Romero (julian@cascabel.ciad.mx)
Mauro E. Valencia (mauro@cascabel.ciad.mx)

Version: 5 Date: 28 June 2007

Author's response to reviews: see over
June the 29th, 2007.

Changes that have done:

Tables have been referred in a numerically consecutive order, table 2 before table 1 (on page 6).

Figures – have been cropped as closely as possible to minimise white space around the image.

The figure legends are before the tables, after the references.

Best regards,
Nayeli Macias.